6.837 Fall 1999 Assignment 1

Find an SGI cluster and free machine

http://www.mit.edu/cview

Register for course on-line at

http://graphics.lcs.mit.edu/classes/6.837/F99/

Online signup, survey, picture

Due by 5pm next Friday, 17 September Required for course registration Create your own readable page at the URL (for uid):

% attach imagery4
% cd /mit/imagery4/6.837/F99/students/uid/WWW
% (edit) homepage.html

(We will validate & create the .../uid/WWW dirs based on today’s signup sheets) Your page should contain:

Your name, and a picture (if you wish) Placeholders for pointers to each assignment A link back to the class page A link to any graphics sites of interest A link to your MIT page (if you wish) Anything else course-related This will produce your page at URL:

http://imagery.mit.edu/imagery4/6.837/
  F99/students/uid/WWW/homepage.html

Check it from your own browser !

Pick up general info sheet on your way out ...

see you on Tuesday.

MIT 6.837 Computer Graphics Thursday 9 September 1999 Page 1